Quantifying the Contribution of Search to Wage Inequality

By Volker Tjaden and Felix Wellschmied*

We empirically establish that one third of job transitions lead to wage losses. Using a quantitative on the job search model we find that 60 percent of them are movements down the job ladder. Accounting for them, our baseline calibration matches the large residual wage inequality in US data while attributing only 13.7 percent of overall wage inequality to the presence of search frictions in the labor market. We can trace the difference between ours and previous much higher estimates to our explicit modeling of non value improving job to job transitions.

JEL: J24, J31, J64

Keywords: Frictional wage dispersion, Search model, Heterogeneity

Mincerian wage regressions explain only about a third of the observed inequality in wage data. Search theoretic models of the labor market offer a compelling explanation for this phenomenon. Their central assumption is that sampling job offers in unemployment takes time and is subject to the opportunity cost of foregone wages. Identical workers; therefore, accept a range of heterogeneous job offers.¹ The literature has come to call this *frictional wage dispersion*. Understanding how much of residual inequality results from search frictions opposed to unobserved worker heterogeneity is of first order importance for judging the efficiency of labor markets and designing appropriate social insurance schemes.

Structural models that seek to answer this question conclude that more than 40 percent of wage inequality within worker skill groups can be explained by the search friction (see Postel-Vinay and Robin (2002) and Carrillo-Tudela (2012)). Hornstein, Krusell and Violante (2012) (henceforth referred to by HKV) show that on the job search is the key mechanism that generates large frictional wage

¹See Mortensen (2003) and the references therein.

^{*} Tjaden: Bonn Graduate School of Economics, Kaiserstrasse 1, D-53113 Bonn, Germany, e-mail: volker.tjaden@uni-bonn.de. Wellschmied: Bonn Graduate School Economics and IZA Bonn, Kaiserstrasse 1, D-53113 Bonn, Germany, e-mail: s3fewell@uni-bonn.de. An earlier version of this paper was entitled "Exploring the Causes of Frictional Wage Inequality". We thank two anonymous referees for detailed suggestions that greatly helped to improve the substance and the presentation of this paper. We also thank Marcus Hagedorn for a very helpful discussion and are grateful for comments from Christian Bayer, Alexander Bick, Jörg Breitung, Carlos Carrillo-Tudela, Wouter Den Haan, Thomas Hintermaier, Philip Jung, Alexander Kriwoluzky, Dirk Krüger, Keith Kuester, Moritz Kuhn, Iourii Manovskii, Monika Merz, Tamás Papp, Petr Sedláček, Konstantinos Tatsiramos, and Gianluca Violante. We also thank seminar participants at the University of Bonn and Pennsylvania and the IHS and conference participants at the 2011 meeting of the Verein für Socialpolitik, the 4th IAB PhD workshop and the 8th ECC/CEPR,IfW Labour Market Workshop. This research has received funding from the European Research Council under the European Union's Seventh Framework Programme (FTP/2007-2013) / ERC Grant agreement no. 282740. Both authors gratefully acknowledge support from the Deutsche Forschungsgemeinschaft (DFG) through the Bonn Graduate School of Economics. Mr Tjaden also gratefully acknowledges a Fulbright grant and thanks the Department of Economics at the University of Pennsylvania for its hospitality.

dispersion. A high offer arrival rate on the job implies that workers are giving up less when moving out of unemployment. This makes them willing to accept relatively poor job offers. Moreover, they quickly move up the job ladder which means a larger share of workers with relatively high wages.

In this paper, we provide evidence from the Survey of Income and Program Participation (SIPP) that an important share of job to job transitions is not value improving. Accounting for this, we calibrate a structural search model with worker and job heterogeneity that replicates observed overall and residual wage inequality. It attributes less than 14 percent of overall wage inequality, or 16 percent of within education group inequality, to the search friction. This result comes in spite of our inclusion of a number of important channels that enlarge the set of acceptable job offers to the worker: skill accumulation on the job, skill loss in unemployment and search on the job. The crucial novelty is the introduction of reallocation shocks that we calibrate to the share of wage losses after a job to job transition. Without them, in a recalibrated model, the variance of the wage offer distribution more than doubles and the contribution of the search friction jumps to over 38 percent, in line with the findings in the previous literature.

The basic intuition for our quantitative results can be summarized in three steps. First, as we demonstrate using a variation of the on the job search model studied by HKV, when all job to job transitions are value improving, workers quickly move into the high ranked jobs from which they are unlikely to accept further offers. Calibrated search efficiency; therefore, has to be high in order to replicate the size of observed job to job flows. This, in turn, means that workers are concentrating in the high ranked jobs even faster. Moreover, because workers give up relatively little search efficiency when accepting employment, they have low reservation wages.

We break this causal chain by introducing what Jolivet, Postel-Vinay and Robin (2006) label a *reallocation shock*: a fraction of the on the job offers leaves the worker only to decide between accepting a random outside offer or moving into unemployment. Workers are more likely to accept in this event than when the alternative is staying with their old job. As a result, they move into high ranked jobs more slowly. Both the inferred overall offer arrival rate on the job and the arrival rate of voluntary offers needed to replicate empirically observed mobility are lower.

Second, keeping the wage offer distribution fixed, wages are less dispersed in the presence of reallocation shocks. They are more compressed at the top because workers move up the job ladder slower. The effect on the reservation wage is a priori ambiguous because reallocation shocks decrease the expected value of high ranked jobs which decreases the reservation wage, while a lower offer arrival rate on the job increases the reservation wage. For realistic calibrations, we find the second effect to dominate which compresses the wage distribution from the bottom.

Third, reallocation shocks lead us to infer a less dispersed wage offer distribu-

tion. We follow Low, Meghir and Pistaferri (2010) in identifying the distribution from the excess variance of wage growth for job switchers relative to job stayers. In the absence of reallocation shocks, many workers hold high value jobs and most job transitions imply small wage improvements such that a high excess variance of wage growth for job switchers can only be rationalized by a very dispersed job offer distribution. In the presence of reallocation shocks, negative wage growth observations and a larger share of acceptable voluntary outside offers mean that the same excess variance of wage growth is consistent with a far less dispersed wage offer distribution. The consequence of a more compressed wage offer distribution is that job effects explain far less of total wage variation.

Can we find evidence for reallocation shocks in the data? Fujita (2011) using data from the UK Labour Force Survey shows that an important share of workers who search on the job do so to avoid unemployment. We extend his analysis using the SIPP employment data to show that reallocation shocks are an important driving force behind observed flows. About a third of all job to job transitions yield lower nominal wages for the worker and neither observable non wage benefits nor higher expected wage growth can account for workers accepting these lower wages. Instead, workers who initially accept a wage cut are more likely to switch jobs again shortly afterwards. Our quantitative model allows us to map the share of losses into the size of reallocation shocks explicitly controlling for measurement error and stochastic innovations to workers' wages. We estimate reallocation shocks to be responsible for 60 percent of observed losses.

The remainder of the paper is structured as follows. Section 1 gives an overview of related literature. In Section 2 we lay out the simple analytical model that highlights the importance of reallocation shocks. Section 3 provides empirical evidence for their presence in the data and highlights stylized facts of residual wage dispersion. We present our full model in Section 4. Section 5 discusses its parameterization. Section 6 presents and analyzes the results, and Section 7 concludes. Additional information on the analytical derivations, the empirical part and the numerical algorithm is relegated to the appendix.²

I. Further Related Literature

Burdett, Carrillo-Tudela and Coles (2011) and Ortego-Marti (2012) show that workers' reservation wages fall significantly in a job ladder model augmented by skill accumulation on the job and skill depreciation in unemployment, respectively. These models match the mean to minimum residual wage in the data, potentially rationalizing all residual inequality as frictional.³ We incorporate these features into our model to give it a fair chance of generating substantial frictional inequality. We show that the inferred job offer distribution provides an upper bound for

²All programs used for data analysis and model solution are available on the authors' web pages.

 $^{^{3}}$ Other recent papers that study conditions under which frictional wage inequality can explain all residual inequality are Papp (2013) and Michelacci, Pijoan-Mas and Ruffo (2012). An earlier example is Bontemps, Robin and van den Berg (2000).

the share of residual inequality that can be thought of as frictional.

Another strand of related literature tries to decompose residual inequality from reduced-form specifications. Abowd, Kramarz and Margolis (1999) and Hagedorn and Manovskii (2010) find that search frictions explain between 7-25 percent of the French inter-industry differential and 6 percent of US wages, respectively. These models rely on exogenous labor mobility and either a permanent component of worker heterogeneity (Abowd, Kramarz and Margolis, 1999), or a stationary shock process (Hagedorn and Manovskii, 2010). Our structural model allows us to explicitly model the selection of workers into matches.⁴ Moreover, we confirm findings from previous studies that residual wage inequality increases strongly over a worker's life-cycle. This suggests a permanent shock component in individual wage potential. Our model allows for such a non stationary shock process and our decomposition of workers' wages over the life-cycle shows that a substantial part of heterogeneity is the result of different employment histories during working life.⁵ Finally, also using the SIPP, Low, Meghir and Pistaferri (2010) use a selection model to infer the wage offer distribution and the shock process of individual wage potential from US wage data. While we ask a different question and use a different empirical strategy, our estimates yield a comparable magnitude for the relative size of idiosyncratic and employment risk.

II. Intuition from a Simple Model

HKV show that the job offer arrival rate on the job is a key parameter determining the wage distribution, and thus the amount of frictional wage inequality, in job ladder models. The higher the on the job offer arrival rate is compared to in unemployment, the smaller is the option value the worker gives up by remaining unemployed and waiting for better offers. Consequently, the minimum wage accepted by workers decreases. Additionally, a high offer arrival rate on the job implies that workers quickly move up the job ladder. This leads to relatively many workers located at high paying jobs. The fact that 1 in 40 employees in the US labor market switches jobs every month seems to hint at high offer arrival rates on the job.

Using an extension to the model studied by HKV, we now demonstrate that one can match high job to job transitions with substantially lower job offer arrival rates when introducing what Jolivet, Postel-Vinay and Robin (2006) label a *reallocation shock*: A fraction of all on the job offers do not allow the worker to stay with his current job, but only leave him to choose between accepting other

 $^{^{4}}$ Abowd, McKinney and Schmutte (2010) discuss that the exogeneity assumption in Abowd, Kramarz and Margolis (1999) is violated because workers sort into jobs with higher match quality. A part of the contribution of this paper; therefore, lies in using additional wage information from job to job transitions to quantify the amount of endogenous upward mobility. We thank an anonymous referee for suggesting this interpretation.

⁵Hagedorn and Manovskii (2010) assume transitory shocks to the worker component and attribute 6 percent of US wage dispersion to search frictions. Using their identification strategy on our non stationary shock process, search frictions explain almost none of the variance of log wages in our simulated data.

employment or becoming unemployed. One may think of these shocks as both transitions within layoff notice period as well as those originating out of non pecuniary motives such as moving in with one's spouse or closer to one's parents.⁶ We show that these shocks crucially affect the wage distribution, both directly and indirectly by the lower inferred on the job offer arrival rate.

Our exposition here is parsimonious and focuses on a few key equations. Appendix A provides a full characterization of the solution. There is a unit mass of homogeneous workers receiving wage offers at Poisson rate λ_u when unemployed and with rate λ when employed. Wage offers are random draws from a cumulative wage offer distribution F(w) with upper support w_{max} that the worker can accept or reject. Time is continuous and workers discount the future at rate r. It is easy to see that the worker follows a reservation wage strategy where the minimum accepted wage is denoted w^* . The asset value of being employed with current wage w is:

$$rW(w) = w + \lambda(1 - \lambda_d) \int_w^{w_{max}} [W(z) - W(w)] dF(z)$$
$$+ \lambda \lambda_d \int_{w^*}^{w_{max}} [W(z) - W(w)] dF(z)$$
$$- (\omega + \lambda \lambda_d F(w^*))(W(w) - U).$$

The worker receives a "normal" on the job offer with probability $\lambda(1 - \lambda_d)$, where λ_d is the probability that an on the job offer is a reallocation shock. The second line is the value of accepting an outside offer after a reallocation shock. Note that now workers accept all wage offers above the reservation wage because they do not have the option to stay with their old jobs. The third line states the value of moving into unemployment which either happens with probability ω after exogenous job destruction, or when the worker refuses an offer after a reallocation shock which occurs with probability $\lambda \lambda_d F(w^*)$. When setting $\lambda_d = 0$, the model reduces to the job ladder model studied by HKV. The asset value of unemployment reads:

$$rU = b + \lambda_u \int_{w^*}^{w_{max}} [W(z) - U] dF(z).$$

An unemployed worker receives benefits b and samples job offers at rate λ_u .

We now establish that a larger share of reallocation shocks decreases the job offer arrival rate inferred from employment transition data and reduces the share of workers with relatively high wages. We then demonstrate that this lowers the amount of wage dispersion implied by the model. The on the job offer arrival rate is typically identified by matching a fixed job to job transition rate, which

 $^{^{6}}$ This is in distinction from a transition where the benefit might have been non monetary but related to the new job like a more permanent work contract or employer provided health insurance.

we label JTJ, and which is given by:

$$JTJ = \lambda(1 - \lambda_d) \underbrace{\int_{w^*}^{w_{max}} [1 - F(z)] dG(z)}_{=:ANO} + \lambda \lambda_d \underbrace{[1 - F(w^*)]}_{=:ARO},$$

where G(w) is the realized distribution of wages. We define ANO as the average probability that a normal on the job offer is accepted and ARO as the probability that an offer is accepted after a reallocation shock. Solving for the implied on the job offer rate gives:

$$\lambda^* = \frac{JTJ}{(1 - \lambda_d)ANO + \lambda_dARO}$$

Increasing the share of reallocation shocks λ_d decreases the inferred on the job offer rate λ^* for two reasons. First, job offers after a reallocation shock are accepted with probability ARO which is larger than the average probability of a normal on the job offer being accepted (ANO). Second, it indirectly affects the latter by changing the wage distribution G(w) which we derive in Appendix A:

(1)
$$G(w) = \frac{F(w) - F(w^*)}{1 - F(w^*)} \underbrace{\underbrace{\omega + \lambda^* \lambda_d}_{\omega + \lambda^* \lambda_d}}_{=:D} \underbrace{\underbrace{\omega + \lambda^* \lambda_d}_{=:C}}_{=:C} [1 - F(w)]^{-1}$$

Reallocation shocks have two effects on the wage distribution. First, like exogenous destruction, they move workers into unemployment from which they subsequently accept any offer above their reservation wage (D). In addition, C shows that they decrease the amount of regular job offers, and thus the speed that workers climb up the job ladder. Consequently, G(w) becomes steeper at low values, i.e., more workers have relatively low wages implying that the probability of a normal offer being accepted (ANO) rises.

In Section V.B, we infer the wage offer distribution F(w) from wage data and show that the mechanisms just outlined have large quantitative implications for the inference. To fix ideas, we here study the effects of changes in λ_d on wage dispersion for a given F(w). HKV propose the ratio of the mean to the minimum wage (Mm-ratio: \bar{w}/w^*) as summary statistic to compare wage dispersion across different classes of search models.⁷ The measure has become a popular statistic in the literature, and for comparability we use it as one summary statistic for wage dispersion later in the paper.

⁷In the models they study, this measure is independent of the wage offer distribution F(w). This does not hold in the environment studies here (see Appendix A for a proof).

In Appendix A, we show that the reservation wage is characterized by:

(2)
$$w^* = b + (\lambda_u - \lambda^*) \int_{w^*}^{w_{max}} \frac{1 - F(z)}{r + \omega + \lambda^* \lambda_d F(w^*) + \lambda^* \lambda_d F(z) + \lambda^* [1 - F(z)]} dz.$$

It is the sum of the flow benefits in unemployment and the option value to keep searching in unemployment. As in a pure job ladder model ($\lambda_d = 0$), the latter is decreasing in the difference $\lambda_u - \lambda$, because workers are giving up less in terms of search efficiency when moving out of unemployment. Similarly, r and ω decrease the value of additional search because workers become more impatient and high wage offers have a lower duration, respectively. Using comparative statics, we demonstrate that changes in λ_d affect the minimum wage directly and indirectly via the implied search efficiency on the job:

$$rac{dw^*}{d\lambda_d} = \underbrace{rac{\partial w^*}{\partial\lambda_d}}_{<0} + \underbrace{rac{\partial w^*}{\partial\lambda^*}rac{\partial\lambda^*}{\partial\lambda_d}}_{>0}.$$

The direct effect of a reallocation shock can be directly read from (2): With probability $F(w^*)$, like exogenous job destruction, it decreases the expected duration of holding employment. Moreover, the further a worker moves up the job ladder, the more likely he will move into a lower ranked job, which decreases the difference in valuation between higher and lower ranked jobs. Both factors decrease the incentive to wait for better offers when moving out of unemployment.⁸ However, the increase in reallocation shocks decreases λ^* which increases the reservation wage. Theoretically, the effect λ_d has on the minimum wage is; therefore, ambiguous and may change depending on parameter values.

The mean wage, is given by:

$$\bar{w} = \int_{w^*}^{w_{max}} w dG(z)$$

Provided our earlier discussion, it should be intuitive that it is a decreasing function of λ_d . More reallocation shocks imply a steeper G(w), and hence a lower mean wage.

For the remainder of this section, to be able to supply graphical representations to our argument, we impose parametric assumptions on the model. Table 1 lists the parameter values. All of them are relatively common in the literature (HKV use similar parameter values in their exposition).

Figure 1 demonstrates how the wage distribution becomes steeper as λ_d increases. Figure 2 shows the drop in the inferred on the job offer arrival rate. The

 $^{^{8}}$ It is this effect which has Hornstein, Krusell and Violante (2007) conclude that reallocation shocks should unambiguously increase the Mm-ratio.



Note: Figure 1 shows the implied distributions of wages paid G(w) for different reallocation shock probabilities λ_d using the parameterization reported in Table 1. Figure 2 reports the implied search efficiency λ for the same exercise, and Figure 3 reports the resulting Mm-ratio.

model estimate reacts particularly sensitive to changes at small values of λ_d . Regarding the reservation wage, Appendix A shows that it rises up to $\lambda_d = 0.35$ and starts to decrease again slowly afterwards. The resulting Mm-ratio from varying λ_d given our parameter values is reported in Figure 3. Especially for low values of λ_d , the Mm-ratio decreases quite sharply in the share of reallocation shocks.

III. Reallocation Shocks and Residual Wage Dispersion in the Data

In this section, we introduce our data set, the Survey of Income and Program Participation (SIPP), and discuss sample selection. We compile different pieces of evidence to show that reallocation shocks are an important feature of the data and link them to existing evidence in other studies. We also obtain the distribution of residual wages from a Mincerian wage regression. Residual inequality is large and shows a substantial increase with worker age.

A. Data Source and Sample Creation

Our analysis requires detailed longitudinal information on wages, worker and job characteristics at a very high temporal resolution. The data set most adequate for these requirements is the SIPP of which we employ the 1993 and 1996

Parameter	Value
b	$0.4\overline{w}$
λ_u	0.3
F(w)	$\ln \mathcal{N}\left(0, 0.04 ight)$
JTJ	2.5 percent
r	0.33 percent

TABLE 1—PARAMETERIZATION SIMPLE MODEL

Note: Unemployment benefits b are a fraction of the mean wage \overline{w} . JTJ designates the job to job transition rate.

panels.⁹ It is a representative sample of the non institutionalized civilian US population maintained by the US Census Bureau.¹⁰ The level of detail it provides in individual records allows us to accurately identify an individual's main job and hourly wages on that job.¹¹ Our initial sample consists of 5,243,222 person/month observations.

Our data cover the years 1993-1995 (1993 sample) and 1996-1999 (1996 sample) providing us with up to 48 months of observations per individual. We use observations from individuals aged 23-55, for whom we require complete information on the individual's employment status, age and employer id. We only consider an individual's primary job¹² and drop workers that are recalled by former employers or have missing reporting months during a job spell.¹³ Moreover, we drop workers reporting to be school enrolled, the self-employed, family-workers, members of the armed forces, workers at non profit companies and anyone whose wage information was imputed by the SIPP.¹⁴ Finally, we truncate the wage distribution at the top and bottom 1 percent to take care of outliers and top-coding.¹⁵ These restrictions leave us with 2,039,345 person/month observations.

We identify job to job transitions as those transitions where the worker works in two consecutive months without reporting unemployment in between,¹⁶ and either the worker's employer identification number or his two-digit occupational identifier changes.¹⁷ Section B of the Web Appendix provides a discussion for alternative measures of job to job transitions and compares our estimate to those obtained from CPS data.

¹¹The survey reports at most two jobs for each 4-month recording period. In case an individual holds more than two jobs, the two jobs with most hours worked are reported.

 12 As primary job we consider the position where the largest share of hours worked is spent.

 13 In case of recall, we choose to exclude those observations because recalled workers likely possess a different search technology than what we include in our model specification.

¹⁴Since our investigation starts from the observation that wage predictions conditional on worker observables explain only a relatively small part of wages, it would seem odd to include wage observations which are mere predictions of these very models

 $^{15}\mathrm{Earnings}$ are topcoded at \$33333 and \$50000 for a four month period in the 1993 and 1996 sample, respectively.

 16 Theoretically, we could use the weekly employment status and count job to job transitions only, when a worker is employed in two consecutive weeks. However, it seems reasonable to assume that a few days in between jobs may be spent on a potential relocation or other pre work sensitivities. Hence, we only discard observations where the worker reports to actively seek a job during non employment.

 17 We think of job to job transitions as a change in the technology operated by the worker; therefore, we include both, changes in job ids (as in Fallick and Fleischman (2004)) and occupation (as in Moscarini and Thomsson (2007)).

 $^{^{9}}$ Our data set is based on CEPR SIPP extracts available for download at $http://www.ceprdata.org/sipp/sipp_data.php$. We modify these abstracts to include further information contained in the SIPP files but not in the original abstracts. Web Appendix A provides additional information on the differences between the two data sets and the steps we take to merge them.

 $^{^{10}{\}rm The}$ 1996 panel oversamples poor households. We use population weights provided by the SIPP throughout our analysis.

AMERICAN ECONOMIC JOURNAL

B. Reallocation Shocks and On the Job Search

This section provides empirical evidence from previous studies and our own data that reallocation shocks are an important feature of employment transitions. While we cannot infer their size directly from the data, Section IV uses a moment from the data together with an extended search model to quantify the share of these shocks.

The existing literature already highlights several shortcomings of a pure job ladder model. Fallick and Fleischman (2004) find for the CPS that a worker who reports to be actively searching on the job is more likely to be unemployed the next month. Fujita (2011) uses a question in the UK labor force survey that asks employees to state a reason for their engaging in on the job search. He finds that of those who report to be actively searching, 12 percent do so for fear of loosing their current job and another 27 percent because they are unsatisfied with their current job due to non pecuniary reasons. Nágypal (2005) shows for a basic job ladder model that the job offer arrival rate on the job has to be higher than during unemployment in order to replicate observed flow rates. Jolivet, Postel-Vinay and Robin (2006) show that in the PSID 23.3 percent of job to job transitions are associated with nominal wage decreases. Including reallocation shocks into a Burdett and Mortensen (1998) model, they find that these shocks account for a third of all job to job offers. Using the SIPP, Connolly and Gottschalk (2008) find that 44.1 percent of all job to job transitions lead to lower real wages. They stress that a higher future expected wage growth may explain initial wage cuts and estimate that 64 percent of male and 81 percent of female wage cuts are truly transitions to lower valued jobs.¹⁸

Regarding our own data, the SIPP asks workers who terminate a job for their reason to do so. The answers further corroborate the evidence previously cited: Only 55 percent of those responding state that they *quit to take another job*. In contrast, 19 percent of jobs ended, because the previous job did not provide the possibility to continue.¹⁹ Adding another 4 percent of cases which pertain to personal or family related issues, this yields up to 23 percent of transitions where, for one reason or another, staying with the old job may not have been an option. There are a number of caveats to the informativeness of this variable: Some of the possible answers are not mutually exclusive, or do not map directly into our interpretation of a reallocation shock. Even more problematic, in less than 30 percent of the cases we identify as job to job transitions, the worker provides an answer.²⁰ ²¹

 $^{^{18}}$ Vice versa, they find that 1.3 percent of females' and 8.6 percent of males' transitions with wage improvements actually go into lower valued matches.

¹⁹This includes the answers on layoff, job was temporary and ended, discharged/fired, employer bankrupt, employer sold business, and slack work or business conditions.

 $^{^{20}}$ For a negligible share the question is not applicable, because only the main job changed, but the worker stays with his old employer. See Web Appendix B for a detailed discussion on how we identify job to job transitions.

²¹Nágypal (2008) discusses the same issue.

Sample Stratification		Share loss	Mean loss
Whole sample		0.344	-0.196
Job characteristics			
	- Non Union to Union	0.346	-0.196
	- Health insurance	0.352	-0.196
	- Education	0.352	-0.196
Old wage			
	- Lowest 25 percent	0.232	-0.16
	- 25-75 percent	0.352	-0.198
	- Top 25 percent	0.457	-0.215

TABLE 2—WAGE CUTS AFTER JOB TO JOB TRANSITIONS

Note: The Table shows the share of workers incurring a cut in nominal hourly wages after a job to job movement for our sample population as a whole as well as for several subsets. Mean loss reports the mean wage loss in log points conditional on suffering a wage cut upon movement. Under *Job characteristics*, the first line excludes workers from the sample who transit from non unionized to unionized jobs, the second and third line additionally exclude workers who move from jobs without health insurance to an employer providing an insurance policy and movements where the new employer subsidizes expenses on education. The panel *Old wage* divides workers based on their wages on the old job. *Source:* Authors' calculations based on SIPP data.

Instead of trying to infer search efficiency from this rather noisy variable, we follow a different strategy in combining employment flow data with accompanying wage dynamics. As we report in Table 2, a pervasive phenomenon in the data are job to job transitions resulting in nominal wage losses. In the whole population, roughly one third of all transitions result in workers earning lower hourly wages in the month after the transition compared to the last month on the previous job.²² Conditional losses are substantial with workers on average receiving about 20 percent lower wages than previously.²³

More than one third of loss-making transitions may seem like a fairly large share at first glance. One possible objection is that wages do not accurately capture the full present value of the new job. As a robustness check, in the segment entitled *Job characteristics*, we exclude transitions from non unionized to unionized jobs since the latter should have higher expected duration and, potentially, higher present value. This does not materially affect our result. Neither does controlling for observable benefit payments such as moving from jobs without health insurance to jobs that provide insurance or into jobs which subsidize education.²⁴

 $^{^{22}}$ As a robustness test, we also constructed three-month-averages of wages before and after a movement to mitigate other sources of reporting error in the months surrounding the transition. This did not affect our estimates.

 $^{^{23}}$ In Web Appendix B, we report the same figures for real wage changes. In that case, the share of loss-making transitions increases to roughly one half with average losses of about 15 percent. In principle, the worker should only consider real wages. But in the presence of some wage rigidity the worker expects a wage loss on his current job as well and compares nominal wages.

 $^{^{24}}$ Given that e.g. Dey and Flinn (2008) show, also using the SIPP, that wages and non wage benefits are positively correlated, this should perhaps not be surprising.

Moreover, losses from job to job transitions are a frequent phenomenon across all segments of the wage distribution from top to bottom as can be seen in the segment *Old wage*. They are twice as likely to occur in the upper quartile of the distribution than in the bottom one, as might be expected given that higher wage earners also have more to lose. Still, even in the bottom part, more than 23 percent of transitions end up in lower paying jobs.

We perform a whole battery of further data stratifications to check whether a particular subgroup or time period is driving the results. Their results are reported in detail in Web Appendix B. Share of losses and conditional changes do not materially change whether we split the sample by year to control for business cycle effects, by gender, age or tenure.²⁵

In Web Appendix B, we also give consideration to an alternative explanation put forward by Postel-Vinay and Robin (2002). They lay out a framework in which workers will accept wage cuts upon job to job transitions, if the option value of working at the other firm is sufficiently high. Indeed, Papp (2013) shows that this framework can rationalize a large amount of wage cuts and large frictional wage dispersion. The key operating mechanism in this class of models is that workers who experienced wage losses have on average steeper observed wage growth afterwards, i.e. wages are backloaded. As we show, there is no indication of that occurring in our data.²⁶

As further piece of evidence that wage losses are the result of transitions into lower ranked jobs, we estimate a probit model conditioning the event of experiencing another subsequent job to job transition on the initial wage change upon movement. Workers who experience a loss making transition are significantly more likely to subsequently transit again. For example, someone having suffered a loss of 20 percent upon movement is 10.3 percent more likely to transit again then someone who experienced an increase of equivalent size and 5.6 percent more likely than someone whose wage remained unchanged.

These different tests lead us to conclude that most of the occurrences of lossmaking transitions are not the result of some benefit not properly accounted for by reported compensation. However, we also cannot not conclude that they all result from reallocation shocks. Simple measurement error in wages is surely part of the story. Shocks to workers' idiosyncratic wage potential may be another contributing factor. In Section IV, we explicitly include these factors in our model specification in order to quantify the amount of reallocation shocks.

 $^{^{25}}$ One exception occurs when we limit our sample to those individuals who report being paid by the hour. In that case, the share of losses drops to 23 percent and conditional losses to 7.8 percent. Still, this figure appears to understate the phenomenon for the population as a whole, because this group is a highly selective subsample of the population with relatively low wages.

 $^{^{26}}$ This appears to contradicts the finding of Connolly and Gottschalk (2008) cited earlier. However, the authors classify wages into only two categories (low,high) and subsequent wage growth into three categories (low,medium,high). In Web Appendix B, we show using a continuous wage growth measure that the data suggest no correlation.

Mm	i-ratio	Mm-ratio by Age Cohort		rt Further measures	
Pctl.		Age	5th Percentile	var. log wages	Gini
1^{st}	3.02	25	1.95		
5^{th}	2.14	36	2.12	0.21	0.29
10^{th}	1.83	49	2.25		

TABLE 3-RESIDUAL WAGE INEQUALITY IN THE 1993/1996 SIPP

Note: The table reports summary measures of residual wage inequality in our data: the mean to minimum ratio, Gini-coefficient and variance of log wages after controlling for worker observables. Since the lowest wage observation in the data is likely the result of measurement error, we report several low percentiles as candidates for the actual minimum wage. Columns 3 to 4 report the Mm-ratio for different age cohorts using the 5th percentile as minimum wage.

Source: Authors' calculations based on SIPP data.

C. Residual Wage Dispersion in the SIPP

Table 3 summarizes measures of residual wage inequality from a regression of log hourly wages²⁷ on a constant, time dummies, a dummy for disabled workers, a dummy for gender, a dummy for marital status, dummies for race (White, Black, Hispanic, Other), dummies for education (Less than high school, High School, Some college, College), 45 regional dummies, the number of kids, experience and experience square. The mean R^2 of this regressions is 0.37 and the variance of log residual wages is 0.21 leaving a significant share of wage variance unexplained.²⁸

The left part of Table 3 summarizes the Mm-ratio in the data. Since the lowest wages are likely the result of measurement error, we report a number of low percentiles as candidate points. Independent of the precise measure, the Mm-ratio, the variance of log wages or the Gini coefficient, residual wage dispersion is large and comparable to previous studies.

While regressions like the one above provide a measure for wage inequality among observationally equivalent workers, it is not clear that this should be interpreted as frictional inequality. Such an interpretation would e.g. falsely assign measurement error and unobserved stochastic innovations to individual wage potential to the search friction. The second column highlights a fact extensively analyzed in the incomplete markets literature, e.g. Storesletten, Telmer and Yaron (2004), but not often addressed in the existing search literature on wage inequality: Cross-sectional residual inequality increases substantially over the life-cycle. Models with a fixed worker wage potential and no on the job search would imply

 $^{^{27}\}mathrm{See}$ Web Appendix A for details on how hourly wages are computed.

²⁸In an earlier version of this paper, we also controlled for unobserved individual worker fixed effect similar to Hornstein, Krusell and Violante (2007). The short observation period of 48 months means that many workers do not experience any job to job transition while they are in the sample. As a result, their individual effect captures the full firm effect in wages and the distribution of residual wages has a large mass point at one. We thank an anonymous referee and Tamás Papp for pointing out this issue to us. Nevertheless, we can compare our model results to this statistic when running the same regression on simulated data. Doing so does not change our conclusions drawn in Section VI.

that inequality does not change with age. Models with on the job search would even predict a decrease in inequality, because workers over time cluster at the higher paying jobs. Therefore, in our model specification, we follow the incomplete markets literature and allow for persistent stochastic innovations to workers' wage potential.

IV. A Quantitative Model of Wage Dispersion

In this section, we extend our simple model studied in Section II by adding worker heterogeneity. We enrich the worker's decision problem by a number of empirically relevant channels that imply larger frictional inequality.²⁹ We also add stochastic innovations to individual wage potential and measurement error in wages which allows us to disentangle wage losses resulting from reallocation shocks from those resulting from other sources.

The model is set in discrete time. Workers differ in their idiosyncratic log wage potential A_t and draw job offers from heterogeneous jobs with log wage contribution Γ .³⁰ When a worker of type A_t and a job of type Γ meet, the wage is given by $w_t = exp(A_t + \Gamma)$.³¹ We assume that search is random, and unemployed workers contact job offers at rate λ_u in which case Γ is drawn from a distribution with cdf $F(\Gamma)$ on support $[\Gamma_m, \Gamma_M]$. Employed workers continue to sample job offers from the same distribution. Following our discussion in Section II, we model some job to job transitions as the result of a reallocation shocks. An employed worker receives a job offer with probability λ and can in general decide to stay with his old match, or form a new one. However, in λ_d of those cases, the outside option becomes unemployment.

Unemployed workers receive unemployment benefits b_t and a value of leisure Z_t that both depend on the worker's idiosyncratic state:

$$b(A_t) = \min\left\{b_{max}, rr_b \cdot \mathbb{E}\left[w_t(A_t, \Gamma)|A_t\right]\right\}$$
$$Z(A_t) = rr_Z \cdot \mathbb{E}\left[w_t(A_t, \Gamma)|A_t\right].$$

where b_{max} are statutory maximum UI payments. Averages are taken over the

²⁹Our focus is on the decision problem of a worker, faces an exogenous job offer distribution. In an earlier version of this paper, Tjaden and Wellschmied (2012), we used a general equilibrium approach with search and matching in the labor market and a Nash-Bargaining game played by workers and firms. We show that the resulting non linear log wage schedule can be almost perfectly approximated by a linear one. For ease of presentation, we opt here for the partial equilibrium representation.

 $^{30}\Gamma$ is the only source of job effects in our model. These can arise from different job specific productivities, match specific effects and, as Winfried Koeniger pointed out to us, differences arising from bargaining over quasi-rents from capital.

³¹Following the existing literature, we assume that wages monotonically increase in the job component conditional on the worker component. Kircher and Eeckhout (2011) and Bagger and Lentz (2012) show that when job effects are independent of match specific effects and the production function has a non zero cross-partial derivative, bargaining models imply a non monotone wage schedule, and a specific sorting of workers over firms is an equilibrium outcome. If this was an important aspect of the data, our model would not control for it.

range of acceptable job offers, which themselves depend on A_t . In the case of unemployment insurance, the dependence on the worker's state capture the fact that benefits are a function of prior contributions and workers with higher wage potential contributed more before becoming unemployed. In the case of the value of leisure, we choose this as the closest analogy to the homogeneous agent world.³²

Workers die with probability ϕ and are replaced by an unemployed labor market entrant whose idiosyncratic log wage potential is drawn from the distribution $N \sim N(\mu_N, \sigma_N^2)$. Burdett, Carrillo-Tudela and Coles (2011) show that introducing experience gains into an on the job search model increases the amount of frictional wage dispersion significantly. To allow for this feature, we let the evolution of workers' wage potential depend on the agent's employment status:

$$A_{t+1} = \begin{cases} A_t + \nu + \epsilon_t & \text{if } employed \\ A_t - \delta + \epsilon_t & \text{if } unemployed. \end{cases}$$

 δ represents skill depreciation while being unemployed and ν represents learning on the job. ϵ is a stochastic shock with $\epsilon \sim N(0, \sigma_{\epsilon}^2)$. We think of shocks to wage potential as demand shocks for specific skills or health shocks. The assumption of a uni-root process in wage potential is in line with most of the labor literature.³³ A non stationary stochastic specification for wages has also become a standard feature of the incomplete markets literature.³⁴ It has so far been less common in quantitative search models.

We summarize the worker problem by the value of employment W and the value of unemployment U. The value of employment depends on a worker's wage potential and a firm's wage contribution, the value of unemployment on the workers' wage potential alone. The value of employment reads:

$$W(A_t, \Gamma) = w_t(A_t, \Gamma) + \beta(1 - \phi) \mathbb{E}_t \{ (1 - \omega) \\ [(1 - \lambda)H + \lambda[(1 - \lambda_d)\Omega_E + \lambda_d\Lambda]] + \omega U(A_{t+1}) \}$$

 \mathbb{E}_t is the expectation operator given all information in period t and ω is an exogenous match destruction shock. For clarity of presentation, we defined the outcome of the choice whether to quit after a bad shock to wage potential as H, the upper envelopes for receiving a regular job offer on the job Ω_E and the upper envelope for receiving a reallocation shock Λ . Let Γ' be the job component at an

 $^{^{32}}$ Furthermore, one can think of this as an, admittedly very stylized, reduced form for capturing wealth heterogeneity. High wage workers tend to have higher asset levels and unemployed workers deplete their assets over time.

 $^{^{33}{\}rm See}$ Abowd and Card (1989), Topel (1991), Topel and Ward (1992), Meghir and Pistaferri (2004) and Low, Meghir and Pistaferri (2010).

 $^{^{34}}$ See for example Krueger et al. (2010).

outside job offer:

$$\begin{split} H &= \max\{W(A_{t+1}, \Gamma), U(A_{t+1})\}\\ \Omega_E &= \int_{\Gamma_m}^{\Gamma_M} \max\{W(A_{t+1}, \Gamma), U(A_{t+1}), W(A_{t+1}, \Gamma')\} dF(\Gamma')\\ \Lambda &= \int_{\Gamma_m}^{\Gamma_M} \max\{W(A_{t+1}, \Gamma'), U(A_{t+1})\} dF(\Gamma'). \end{split}$$

The value of unemployment solves:

$$U(A_t) = b(A_t) + Z(A_t) + \beta(1 - \phi) \mathbb{E}_t \{ (1 - \lambda_u) U(A_{t+1}) + \lambda_u \int_{\Gamma_m}^{\Gamma_M} \max\{ W(A_{t+1}, \Gamma), U(A_{t+1}) \} dF(\Gamma) \}.$$

V. Parameterization

This section proceeds as follows: We first discuss our calibration regarding non distributional parameters (preferences, institutions, flow rates) in Section V.A. In Section V.B, we discuss our calibration of distributional parameters. Table 4 summarizes our calibration.

A. Non Distributional Parameters

The model period is one month. When comparing monthly wages in the model to hourly wages in the data, we assume an average of 160 work hours per month. The length of a period is of importance, because it puts an upper bound on the job offer probability λ_u and the minimum duration of an unemployment spell. A maximum of one offer per month is well supported by the data,³⁵ but the second constraint is likely to be binding.³⁶

We calculate the employment to unemployment and unemployment to employment flow rates in our SIPP sample. The exogenous job destruction rate ω is set such that the total job destruction rate, the sum of endogenous and exogenous movements from employment to unemployment, is 0.65 percent per month. We attach to λ_u a value that implies a monthly job finding rate of 12.3 percent.

Information on job to job movements and accompanying wage changes identify λ and λ_d . We adjust λ to imply that 1.43 percent of workers switch employers every period. Our identifying assumption for separating voluntary and involuntary movements is that voluntary movements always result in expected wage increases.

16

 $^{^{35}}$ Holzer (1988) reports based on NLSY data that 34 percent of the unemployed received at least one job offer and 12 percent received more than one offer per month.

³⁶See Clark and Summers (1979). Our model cannot by construction match the high observed outflow rates within the first month. However, time disaggregation below one month is rather costly, because our numerical algorithm uses value function iteration, which converges at a rate of β .

Together with the losses due to stochastic idiosyncratic shocks to wage potential and measurement error, both of which are calibrated below, setting λ_d to 0.1 allows us to replicate that 34 percent of job to job movements result in nominal wage losses.³⁷

The flow rates estimated from our sample are considerably lower than comparable estimates commonly found in the CPS. In Web Appendix A, we discuss that this is largely explained by fact that our sample selection criteria lead us to focus on individuals with relatively stable employment histories. Estimated flow rates from our raw sample are considerably larger and comparable to those found in the CPS.³⁸

Consistent with findings from Siegel (2002) for average bond and stock returns, we set β to imply a yearly interest rate of 4 percent. Next, we consider the flow value of unemployment. We set the replacement rate rr_b to 25 percent. As argued in Hall and Milgrom (2008) this provides a parsimonious description of the system. The maximum UI benefit payment is set to 1168 \$, which is the average across US states. The parameter determining the value of leisure rr_z is set to 15 percent which yields a total replacement rate of 40 percent when entering into unemployment as in Shimer (2005).³⁹

We choose an indirect inference approach in calibrating experience and depreciation.⁴⁰ In the data, we regress log hourly wages at zero tenure on individual fixed effects, time fixed effects and a quadratic polynomial in experience. The regression yields an average increase in annual wages of 3 percent per year of experience over a working life of 25 years.⁴¹ We then use our model solution to simulate 30000 worker histories and draw a panel of the same length as the SIPP. We perform a similar regression⁴² in our simulated data to control for selection and adjust ν to match this statistic. For skill depreciation δ we run a regression of log hourly wages after an unemployment to employment transition on the duration of the previous unemployment spell and worker observables. The results imply that an extra month of unemployment reduces wages by 0.39 percent. We then again replicate this regression in our data and adjust δ to match the

³⁷The share of realized job to job transitions that result from a reallocation shock is 28 percent, which compares nicely with our survey evidence presented in Section III.B. In total, 60 percent of loss making transitions result from reallocation shocks. Our explicit modeling of measurement error and shocks to individual wage potential decrease the estimate of reallocation shocks considerably compared to the studies of Jolivet, Postel-Vinay and Robin (2006) and Connolly and Gottschalk (2008).

 $^{^{38}}$ Moreover, equation (2) highlights that for a worker's decision problem only the difference between the on and off the job offer arrival rates matters. Both are significantly lower in our study compared to the ones reported by e.g., Fallick and Fleischman (2004) based on CPS data, but the difference has a comparable size.

³⁹The value of leisure is a much discussed object in the literature and Hall and Milgrom (2008) suggest a total replacement rate of 0.71. In Web Appendix D we show that using this higher rate leaves our results virtually unaffected.

⁴⁰We thank an anonymous referee for suggesting this approach to us.

⁴¹Altonji and Williams (1998) report very similar results.

 $^{^{42}}$ Experience is imperfectly measured in the SIPP. Workers are asked how many years they worker at least 6 full months since first entering the labor market. We construct the same measure for yearly experience in our simulated data.

regression statistic.

B. Distributional Parameters

We now describe the way we calibrate the variance of the wage offer distribution σ_F^2 , idiosyncratic shocks to wage potential σ_ϵ^2 , initial worker dispersion σ_N^2 and the measurement error process. None of the statistics is directly observable in the data because observed wages at all stages of the life-cycle are a function of all three factors. Moreover, workers endogenously select themselves into and out of employment and into employment with jobs of specific wage offers in response to idiosyncratic productivity developments. Instead, we identify them from within our model by jointly calibrating them together with all other model parameters.

MEASURING JOB HETEROGENEITY. — Similar to Low, Meghir and Pistaferri (2010), our identification of the job offer distribution rests on the excess variance of job switchers and job stayers in the data. Other than specifying an additive specification for log wages and assuming the firm contribution to be log normally distributed, this identification only relies on the assumption that measurement error for job switchers is the same as for job stayers. Web Appendix C provides evidence for this assumption.

In our SIPP data, we assume that wages are generated by:

(3)
$$ln(w_{i,t}) = \alpha_0 + \alpha_1 d_t + \alpha_2 \mathbf{Z}_i + \beta_2 \Gamma_i + e_{i,t}$$

where d_t captures aggregate states, such as TFP and \mathbf{Z}_i is a vector of idiosyncratic components. We split the unobservable $e_{i,t}$ into two parts:

$$e_{i,t} = r_{i,t} + A_{i,t}$$

Like in the model $A_{i,t}$ is assumed to follow a random walk with drift and innovations $\epsilon_{i,t}$, and $r_{i,t}$ captures measurement error. For our present purpose, we have to make no further assumptions regarding the distributional properties of measurement error.

First-differencing eliminates the idiosyncratic wage components. As mentioned above, we only observe a self-selected subset of the realizations of Γ and ϵ as agents can quit into unemployment after negative idiosyncratic shocks and refuse wage offers. The subsets of observed realizations Γ^{obs} and ϵ^{obs} are themselves random variables which follow distributions of unknown functional forms. However, we can use the workers' decision rules, which determine for each (A_t, Γ) combination whether to form or continue a match, to map these moments back into the structural parameters.

Define observed wage growth when a job to job transition takes place

$$\Delta ln(w_{i,t}^b) = \nu + \kappa_t + [\Gamma_i^{obs} - \Gamma_{i-1}^{obs}] + \epsilon_{i,t}^{obs} + \Delta r_{i,t}$$

Variable	Target
$\beta = 0.997$	4 percent annual interest rate
$rr_b = 0.25$	$\frac{b_{mean}}{w_{max}} = 0.25$
$rr_{Z} = 0.15$	$\frac{Z_{mean}}{w_{mean}} = 0.15$
b_{max}	1168\$
$\omega = 6.5 * 10^{-3}$	EU flow rate of 0.0065
$\lambda_u = 0.124$	UE flow rate of 0.123
$\lambda = 0.043$	JTJ flow rate of 0.0147
$\lambda_d = 0.096$	34 percent of wage cuts upon JTJ movements
$\nu = 2.5 * 10^{-3}$	3 percent yearly experience coefficient
$\delta = 2.3 * 10^{-3}$	0.39 percent monthly depreciation coefficient
$\phi = 0.04$	33 years of working life
$\sigma_F = 0.163, \Gamma \sim N(0, \sigma_F^2)$	Equation $(4)=0.0397$
$\sigma_{\epsilon} = 0.016, \ \epsilon \sim N(0, \sigma_{\epsilon}^2)$	Life-cycle wage profile
$\sigma_N = 0.293, N \sim N(\mu_N, \sigma_N^2)$	Life-cycle wage profile
$\sigma_{\iota} = 0.119, \ \iota \sim N(0, \sigma_{\iota}^2)$	Estimation
$\mu_N = 5.618$	Mean monthly wage 2139\$

TABLE 4—CALIBRATION

Note: The left column states the calibrated variable with its value and the second states the relevant moment. EU stands for employment to unemployment, UE for unemployment to employment, and JTJ for job to job.

and when no such transition takes place

$$\Delta ln(w_{i,t}^w) = \nu + \kappa_t + \epsilon_{i,t}^{obs} + \Delta r_{i,t}$$

where $\kappa_t = \alpha_1 (d_t - d_{t-1})$. After regressing out a constant and time dummies, we obtain the residual excess variance of job movers relative to job stayers:⁴³

(4)
$$Var\left[\Delta ln(\hat{w}_{i,t}^{b})\right] - Var\left[\Delta ln(\hat{w}_{i,t}^{w})\right] = Var\left[\Gamma_{i}^{obs} - \Gamma_{i,-1}^{obs}\right] + Cov\left[\epsilon_{i,t}^{obs}(\Gamma_{i}^{obs} - \Gamma_{i,-1}^{obs})\right]$$

where we have invoked the assumption that measurement error is uncorrelated with the event of job switching.

Equation (4) also holds in our model and we use it as a calibration target for σ_F^2 . The endogenous sorting that causes the observed distribution in the data to differ from the true one is also present in our model.

 $^{^{43}}$ We delete the top and bottom 0.5 percent of the wage growth observations to get rid of reporting error.

CALIBRATING IDIOSYNCRATIC WAGE POTENTIAL. — Similar to Storesletten, Telmer and Yaron (2004), we calibrate the variance of idiosyncratic wage shocks to the life-cycle profile of cross sectional residual wage dispersion.⁴⁴ While we explicitly model initial worker heterogeneity and experience gains, the data possesses wellknown idiosyncratic wage components absent from our model that we regress out (gender, race, marriage, number of children, disability and time dummies).⁴⁵ We then choose σ_N^2 to match the initial variance of residual log wage inequality not explained by job effects and σ_{ϵ}^2 to match its increase over the life cycle.

Lastly, wage fluctuations may result from measurement error. To accurately identify the share of reallocation shocks and to properly calibrate the innovations to individual wage potential, we require an explicit treatment for this source of wage fluctuations. At this point, we need to make further assumptions regarding its statistical properties. Web Appendix C shows that the autocovariance function of within job wage growth goes to zero at longer lags. Therefore, we follow Meghir and Pistaferri (2004) and postulate an MA(q) process (i.e. $r_{i,t} = \Theta(q)\iota_{i,t} =$ $\iota_{i,t} - \sum_{j=1}^{q} \theta_{j}\iota_{i,t-j}$). The autocovariance function is close to zero after 12 lags, such that we fix q at 12. Assuming $\mathbb{E}(\epsilon_{i,t}^{obs}\epsilon_{i,t-j}^{obs}) = 0 \forall j \neq 0$, we obtain the parameters $\Theta(12)$ and σ_{ι} using Maximum Likelihood estimation and Kalman filtering.⁴⁶ Web Appendix C supplies further detail on the procedure and shows that θ_{12} is indeed estimated close to zero.

VI. Results

We now present the main results of our paper. In Section VI.A we demonstrate that our model generates residual wage dispersion of the size estimated in the data and that it matches its life-cycle profile. Moreover, the model provides a close fit to the shape of the overall wage distribution. Section VI.B discusses the structurally inferred parameters of the wage offer distribution and of idiosyncratic wage uncertainty. We then go on to determine the relative contributions of job dispersion, development in workers' wage potential and the distribution of workers over jobs to overall wage dispersion. Our results attribute 13.7 percent of wage inequality to the presence of the search friction. Using an alternative model without reallocation shocks, the estimate jumps up to the size previously estimated in the data.

⁴⁴In principle, we could derive a moment condition similar to the one above to identify idiosyncratic wage uncertainty (see Meghir and Pistaferri (2004) for more details). Whereas the identification of the job component only required two consecutive wage observations, the maximum spell length of 48 months in the SIPP now becomes more of on an issue which is why we opt for a different approach.

⁴⁵We purify our data of these effects, which are well-known drivers of wages, because we think them inadequately represented by our model set-up. Gender and race biases are likely the result of discrimination. Marriage stands in for a joint labor supply decision absent from our model. Disability and the number of children likely do represent productivity, but not in a way adequately captured by our model. ⁴⁶We thank Johannes Pfeifer for providing the Kalman filtering routine to us.



FIGURE 4. MM-RATIO OVER THE LIFE-CYCLE FIG

FIGURE 5. WAGE DISTRIBUTIONS

Note: Figure 4 plots the Mm-ratio by age in the model against the data. Figure 5 compares demeaned density functions of wages after applying a kernel smoother. *Source:* Authors' calculations based on model simulation and SIPP data.

A. Empirical Fit

We simulate a cohort of 30000 workers over their life-cycle. From the resulting individual paths we sample 48 month observation spells to generate a data set of the same length as the SIPP. We then run a regression of log wages on a constant and experience to calculate the model counterpart to our measure of residual wages in the data. Table 5 summarizes our results.

	Mean-N	Min Ratio)	Gi	ni	Var(log	$g(\tilde{w}_{it}))$
		Model	Data	Model	Data	Model	Data
	1^{st}	3.01	3.02				
Pctl.	5^{th}	2.21	2.14	0.24	0.29	0.18	0.21
	10^{th}	1.89	1.83				

TABLE 5—RESIDUAL WAGE DISPERSION

Note: The table compares the size of the residual wage dispersion generated by our baseline specification to the one found in the 1993/1996 SIPP. The first two columns report the Mm-ratio in the model and the data using the 1st, 5th, and 10th percentile as possible minimum wages. As further summary statistics, we compare the Gini coefficient and the variance of log wages.

Source: Authors' calculations based on model simulation and SIPP data.

The mean residual wage paid is 3.01 times the smallest observation evaluated at the first percentile. When looking at higher percentiles, model and data line up closely as well. Other summary statistics of inequality also indicate a good fit: the Gini coefficient and the variance of residual log wages are slightly smaller, but close to those found in our data set.

In Section III.C, we discussed that a characteristic feature of residual inequality

Specification	σ_F	σ_ϵ	σ_N	λ
Baseline	0.163	0.016	0.293	0.043
job ladder model ($\lambda_d = 0$)	0.296	0.017	0.117	0.1

TABLE 6—WAGE OFFER DISTRIBUTION AND IDIOSYNCRATIC RISK

is its increase over the life-cycle and used the fact to motivate our stochastic wage potential process. Figure 4 compares the model to the data along that dimension. We closely match the magnitude of the increase over the life-cycle, while missing the concave shape at the end.

In our subsequent analysis, we use our model to compute the contribution of search induced wage inequality to overall wage inequality in the population crosssection. Therefore, we need to verify that our model fits the data along that dimension. As discussed previously, there are a few well-known wage determinants in the data that our model is not designed to include. In what follows, we first regress log wages in our data on a constant and dummies for disability, gender, marriage status, the number of kids, time and race. These factors account for 13.3 percent of log wage variation. We compare the wage distribution from our model to the resulting distribution. Figure 5 plots the kernel estimator of the density function of wages after transforming the data back to levels against its model counterpart. The two graphs match up almost perfectly well. There is substantial inequality and the distribution features the characteristic right skew.

B. Underlying Sources of Inequality

Confident that our model features the main determinants of wage inequality, we use it to infer the relative importance of differing initial abilities (σ_N , in our model), uncertainty of idiosyncratic wage potential (σ_ϵ), the search friction (σ_F) and a sorting term to be introduced below in explaining overall wage inequality. Our calibrated parameters are displayed in the first line of Table 6.

Our model implies a direct link between observed wage outcomes and these deep parameters. In order to map it out, we use our simulated data and consider the following variance decomposition, which we separately estimate for each age group in our simulated data

$$Var(ln(w_i)) = Var(A_i) + Var(\Gamma_i) + 2Cov(A_i, \Gamma_i) + Var(r_i).$$

The left panel of Figure 6 illustrates the results. For young workers, job heterogeneity explains about 24 percent of overall log wage variance but that number drops as workers' employment histories become more diverse. Our model identi-

Note: The table displays the standard deviations of the wage offer distribution and of the idiosyncratic wage shock. The first line refers to the baseline specification and the second one to a calibration of a "pure" job ladder model.



FIGURE 6. CONTRIBUTION OF SEARCH FRICTIONS TO OVERALL WAGE DISPERSION BASELINE V. JOB LADDER MODEL

fies worker heterogeneity as the dominant factor in explaining variations in wages and this effect is increasing in age. Measurement error is responsible for about 2.4 percent of variation. Sorting of workers over job types has a mild positive effect. In a population weighted average, frictional wage dispersion accounts for 15.5 percent of wage inequality within our model. Given that we eliminated 13.3 percent of wage variation through our fixed effect regression, this implies frictional inequality to account for 13.7 percent of overall wage inequality present in our data.

C. On the Job Search and Structural Inference

Previous estimates from structural search models that match overall wage inequality imply a much larger role for frictional inequality than we do. After controlling for observable worker skills, Postel-Vinay and Robin (2002) suggest numbers up to 50 percent and Carrillo-Tudela (2012) reports estimates around 40 percent. Even when controlling for education, which explains about 15 percent of wage variation in our data, our model attributes only 16 percent of the within group inequality to the search friction. In this section, we investigate whether the introduction of the reallocation shock alone can explain the large quantitative discrepancy. We also highlight how the mechanisms outlined in Section II interact when we identify the variance of the job offer distribution.

We re-calibrate our baseline model to a more common job ladder model setting, $\lambda_d = 0$, and neglect wage losses upon transition as calibration target. With a Mmratio of 3.45 at the first percentile, the model yields a residual inequality of similar size as our baseline specification. To demonstrate that measurement error and

Note: The graphs display the cumulative contribution of sorting (black area), firm effects (dark grey area) measurement error(medium grey area) and worker heterogeneity (light area) to the variance of log wages, conditional on age. The left panel is from our baseline specification, the right panel results from a job ladder model with idiosyncratic productivity risk.

stochastic wages alone cannot account for the stylized facts outlined in Section III.B, we compare moments of wage dynamics upon job to job movement in the data to our our baseline specification and the job ladder-model. Table 7 displays the results.

In the data, job to job movements on average result in wage gains of 3.3 percent. Conditional on suffering a wage loss upon movement, workers lose 19.6 percent of their previous wages. Our baseline specification fares quite well in reproducing these statistics. Wage gains are too high, but the order of magnitude is comparable. The model does well in reproducing the large conditional wage losses. In Web Appendix D, we show that our baseline specification is also in line with the large initial wage gains at job to job transitions reported by Topel and Ward (1992) and the convex decrease of these gains over experience. In the pure job ladder model, average wage gains at job to job transitions of 23 percent are much too large compared to the data. Since workers in this model only transit to higher ranked jobs, the wage losses are only observed as result of a negative shock to individual wage potential or due to measurement error. A conditional 9 percent average wage loss clearly fails in this respect. We come back to this fact below.

We now investigate what these differences imply for the inferred importance of difference sources of wage inequality. The right panel of Figure 6 shows that this model paints a much changed picture of the different sources of wage inequality, when compared to our baseline specification. The cross-sectional average for the contribution of frictional wage dispersion more than doubles to about 44 percent (38.8 percent of wage variation in the data) with values as high as 78 percent for the youngest workers. Closely related is an almost doubling in the inferred standard deviation of the wage offer distribution as can be seen in the second row of Table 6.

The reason for these results can be traced back to the role of reallocation shocks. Section II demonstrated that in the absence of reallocation shocks, the inferred job offer arrival rate on the job is higher and more workers are in the right tail of the job offer distribution. Table 6 shows that our recalibrated model implies an on the job offer arrival rate more than twice as large as our baseline calibration.

Specification	Avg. gain	Avg. loss
Data	0.033	-0.196
Baseline	0.071	-0.186

0.227

-0.09

TABLE 7—WAGE CHANGES FROM JOB TO JOB MOVEMENTS

Note: The table compares the model baseline specification with a pure on the job search version in their implications for job to job transitions. Statistics are the resulting average wage gain upon job movement and the average wage loss, conditional on observing a loss. *Data* refers to computation from the SIPP for nominal wages.

Source: Authors' calculations based on model simulation and SIPP data.

job ladder model ($\lambda_d = 0$)

=

25

Consequently, workers quickly move into very high ranked matches, accept further outside offers only infrequently and wage improvements are relatively small. Since they also do not experience large losses when moving, the implied wage offer distribution has to spread out substantially to reproduce the observed excess variance for job switchers.⁴⁷ On the flip side, most initial dispersion is explained by job effects and the inferred initial worker heterogeneity drops by half in terms of its standard deviation. The two model versions tell rather different stories about the sources of life-time wage inequality. As a robustness analysis, we decrease the share of reallocation shocks exogenously by a half. Results of this exercise are reported in Web Appendix D. The variance decomposition yields results close to our baseline case, showing that already some reallocation shocks overturn the strong implications from the pure job ladder model.

VII. Conclusion

We solve a rich structural model of job and worker heterogeneity to quantify the importance of the search friction in generating wage inequality. Our model features several major channels that expand the range of acceptable offers to the workers creating larger *frictional* inequality: skill accumulation on the job, skill loss in unemployment and search on the job. The baseline calibration reproduces both overall and residual wage inequality. Nonetheless, the search friction accounts for only 13.7 percent of total inequality.

The large quantitative difference to previous estimates stems from our introduction of reallocation shocks upon job to job transitions. These shocks allow our model to match a large job to job transition rate in the data with a relatively low on the job offer arrival rate. As a consequence, the endogenous wage distribution features few workers at high ranked jobs. The calibrated variance of the job offer distribution is relatively small and only a small share of wage variation can be explained by job differences.

Empirically, we provide various pieces of evidence to show that reallocation shocks provide a fitting description for about a quarter of observed job to job transitions. Most importantly, about one third of all job to job transitions end up with lower nominal wages than on the previous job. This finding is robust to both controlling for observed benefit payments as well as all kinds of data stratification.

 $^{^{47}}$ As rightfully noted by a referee, a higher offer arrival rate on the job lowers the reservation wage. This in turn may lead to a larger excess variance of wage growth of job switchers. However, we find across different calibrations that this effect is never dominant.

SOLVING THE MODEL OF SECTION II

This section derives implicit solutions for the minimum wage, the mean wage, the wage distribution and the relationship between job to job transitions and the job offer rate for the model presented in Section II.

Recall the worker problem:

$$\begin{split} rW(w) &= w + \lambda(1 - \lambda_d) \int_w^{w_{max}} [W(z) - W(w)] dF(z) \\ &+ \lambda \lambda_d \int_{w^*}^{w_{max}} [W(z) - W(w)] dF(z) \\ &- (\omega + \lambda \lambda_d F(w^*)) (W(w) - U) \\ rU &= b + \lambda_u \int_{w^*}^{w_{max}} [W(z) - U] dF(z), \end{split}$$

where F(w) is the cdf of the wage offer distribution with upper support w_{max} , λ is the job offer arrival rate on the job, λ_d is the share of reallocation shocks, ω is the job destruction rate and λ_u the job offer arrival rate during unemployment. Evaluating the asset value of employment at w^* and setting it equal to the asset value of unemployment yields:

$$w^* = b + (\lambda_u - \lambda) \int_{w^*}^{w_{max}} W'(z) [1 - F(z)] dz.$$

Differentiating the asset value of employment with respect to w yields

$$W'(w) = \frac{1}{\omega + \lambda \lambda_d F(w^*) + r + \lambda \lambda_d + \lambda (1 - \lambda_d) [1 - F(w)]}$$

Therefore, we obtain an implicit solution for the reservation wage reported in Section II:

(A1)
$$w^* = b + (\lambda_u - \lambda) \int_{w^*}^{w_{max}} \frac{1 - F(z)}{r + \omega + \lambda \lambda_d F(w^*) + \lambda \lambda_d F(z) + \lambda [1 - F(z)]} dz.$$

Figure A1 highlights the non monotone relationship between λ_d and w^* discussed in Section II.

We now derive an implicit solution for the wage distribution G(w). A stationary distribution of employment over wages implies:

(A2)
$$(1-u)G(w)[\omega + \lambda\lambda_d F(w^*) + \lambda[1-F(w)]] = u\lambda_u[F(w) - F(w^*)] + (1-u)\lambda\lambda_d[1-G(w)][F(w) - F(w^*)]$$

26



Note: The figure displays the relationship between the share of reallocation shocks, λ_d , the minimum wage and the mean wage for the calibration performed in Section II.

Rearranging yields

$$G(w) = \frac{u\lambda_u + (1-u)\lambda\lambda_d}{1-u} \frac{F(w) - F(w^*)}{\omega + \lambda[1-F(w)] + \lambda\lambda_d F(w)}$$

Evaluating (A2) at w^{max} yields

$$\frac{u}{1-u} = \frac{\omega + \lambda \lambda_d F(w^*)}{\lambda_u [1 - F(w^*)]}.$$

Substituting into (A2) gives the solution for G(w):

(A3)
$$G(w) = \frac{F(w) - F(w^*)}{1 - F(w^*)} \frac{\omega + \lambda \lambda_d}{\omega + \lambda \lambda_d F(w) + \lambda [1 - F(w)]}$$

We now derive an implicit solution for the relationship between λ and the job to job transition rate that we omit in the main paper for parsimony. Total job to job flows are given by:

$$JTJ = \lambda \lambda_d [1 - F(w^*)] + \lambda (1 - \lambda_d) \int_{w^*}^{w_{max}} [1 - F(z)] dG(z).$$

Integrating the equation by parts yields

$$JTJ = \lambda \lambda_d [1 - F(w^*)] + \lambda (1 - \lambda_d) \int_{w^*}^{w_{max}} G(z) dF(z)$$

Substituting in G(w) gives

$$JTJ = \lambda \lambda_d [1 - F(w^*)] + \lambda (1 - \lambda_d) \frac{\omega + \lambda \lambda_d}{1 - F(w^*)} \int_{w^*}^{w_{max}} \frac{F(z) - F(w^*)}{\omega + \lambda \lambda_d + \lambda (1 - \lambda_d) [1 - F(z)]} dF(z).$$

Replace z = F(z) to obtain

(A4)
$$JTJ = \lambda \lambda_d [1 - F(w^*)] + \lambda (1 - \lambda_d) \frac{\omega + \lambda \lambda_d}{1 - F(w^*)} \int_{F(w^*)}^1 \frac{z - F(w^*)}{\omega + \lambda \lambda_d + \lambda (1 - \lambda_d)[1 - z]} dz.$$

Solving the integral yields:

$$\begin{split} \int_{F(w^*)}^1 \frac{z - F(w^*)}{\omega + \lambda \lambda_d + \lambda (1 - \lambda_d)[1 - z]} dz = \\ & \Big| - \frac{\lambda (1 - \lambda_d) z + [\omega + \lambda] log(\omega + \lambda \lambda_d + \lambda (1 - \lambda_d)[1 - z])}{[\lambda (1 - \lambda_d)]^2} \\ & + \frac{F(w^*) log(\omega + \lambda \lambda_d + \lambda (1 - \lambda_d)[1 - z])}{\lambda (1 - \lambda_d)} \Big|_{F(w^*)}^1. \end{split}$$

Finally, we can derive a solution for the mean wage:

$$\bar{w} = \int_{w^*}^{w_{max}} w dG(z).$$

Integration by parts yields

$$\begin{split} \bar{w} &= w^{max} - \int_{w^*}^{w_{max}} G(z) dz \\ &= [w^{max} - w^*] + w^* - \int_{w^*}^{w_{max}} G(z) dz \\ &= w^* + \int_{w^*}^{w_{max}} [1 - G(z)] dz \\ &= w^* + \frac{\omega + \lambda - \lambda(1 - \lambda_d) F(w^*)}{1 - F(w^*)} \int_{w^*}^{w_{max}} \frac{1 - F(z)}{\omega + \lambda \lambda_d + \lambda(1 - \lambda_d) [1 - F(z)]} dz, \end{split}$$

which is an implicit solution for \bar{w} . Figure A2 shows the resulting downward sloping relationship between λ_d and λ . Upon inspection to the mean and minimum wage, it becomes apparent that their ratio is not a moment independent of F(w)in our model with reallocation shocks.

*

REFERENCES

- Abowd, John M., and David Card. 1989. "On the Covariance Structure of Earnings and Hours Changes." *Econometrica*, 57(2): 411–445.
- Abowd, John M., Francis Kramarz, and David N. Margolis. 1999. "High Wage Workers and High Wage Firms." *Econometrica*, 67(2): 251–334.
- Abowd, John M., Kevin McKinney, and Ian M. Schmutte. 2010. "How Important is Endogenous Mobility for Measuring Employer and Employee Heterogeneity?" http://www.sole-jole.org/11332.pdf.
- Altonji, Joseph G., and Nicolas Williams. 1998. "The Effects of Labor Market Experience, Job Seniority, and job Mobility on Wages." *Research in Labor Economics*, 17(1): 233–276.
- Bagger, Jesper, and Rasmus Lentz. 2012. "An Equilibrium Model of Wage Dispersion and Sorting." https://sites.google.com/a/wisc.edu/rasmus-lentz/research.
- Bontemps, Christian, Jean-Marc Robin, and Gerard J. van den Berg. 2000. "Equilibrium Search with Continuous Productivity Dispersion: Theory and Nonparametric Estimation." *International Economic Review*, 41(2): 305–358.
- Burdett, Kenneth, and Dale T. Mortensen. 1998. "Wage Differentials, Employer Size, and Unemployment." *International Economic Review*, 39(2): 257–273.
- Burdett, Kenneth, Carlos Carrillo-Tudela, and Melvyn Coles. 2011. "Human Capital Accumulation and Labour Market Equilibrium." International Economic Review, 52(3): 657–677.
- Carrillo-Tudela, Carlos. 2012. "Job Search, Human Capital and Wage Inequality." IZA Working Paper No. 6949.
- Center for Economic and Policy Research. 2010. "SIPP Uniform Extracts, Version 2.1.5." (accessed March 10th, 2011).
- Clark, Kim B., and Lawrence H. Summers. 1979. "Labor Market Dynamics and Unemployment: A Reconsideration." *Brookings Papers on Economic Activity*, 10(1): 13–72.
- Connolly, Helen, and Peter Gottschalk. 2008. "Wage Cuts as Investment in Future Wage Growth." *Labour*, 22(1): 1–22.

- Dey, Matthew S, and Christopher J Flinn. 2008. "Household Search and Health Insurance Coverage." Journal of Econometrics, 145(1-2): 43–63.
- Fallick, Bruce, and Charles A. Fleischman. 2004. "Employer-to-employer flows in the U.S. labor market: the complete picture of gross worker flows." Board of Governors of the Federal Reserve System (U.S.) Finance and Economics Discussion Series 2004-34.
- **Fujita, Shigeru.** 2011. "Reality of on-the-job search." Federal Reserve Bank of Philadelphia Working Paper 10-34/R.
- Hagedorn, Marcus, and Iourii Manovskii. 2010. "Search Frictions and Wage Dispersion." http://economics.sas.upenn.edu/~manovski/research.html.
- Hall, Robert E., and Paul R. Milgrom. 2008. "The Limited Influence of Unemployment on the Wage Bargain." *American Economic Review*, 98(4): 1653– 1674.
- Holzer, Harry J. 1988. "Search Method Use by Unemployed Youth." *Journal* of Labor Economics, 6(1): 1–20.
- Hornstein, Andreas, Per Krusell, and Giovanni L. Violante. 2007. "Frictional Wage Dispersion in Search Models: A Quantitative Assessment." National Bureau of Economic Research, Inc NBER Working Papers 13674.
- Hornstein, Andreas, Per Krusell, and Giovanni L. Violante. 2012. "Frictional Wage Dispersion in Search Models: A Quantitative Assessment." American Economic Review, 101(7): 2873–2898.
- Jolivet, Gregory, Fabien Postel-Vinay, and Jean-Marc Robin. 2006. "The empirical content of the job search model: Labor mobility and wage distributions in Europe and the US." *European Economic Review*, 50(4): 877–907.
- Kircher, Philipp A., and Jan Eeckhout. 2011. "Identifying Sorting-In Theory." The Review of Economic Studies, 78(3): 872–906.
- Krueger, Dirk, Fabrizio Perri, Luigi Pistaferri, and Giovanni L. Violante. 2010. "Cross Sectional Facts for Macroeconomists." *Review of Economic Dynamics*, 13(1): 1–14.
- Low, Hamish, Costas Meghir, and Luigi Pistaferri. 2010. "Wage Risk and Employment Risk over the Life Cycle." *American Economic Review*, 100(4): 1432–1467.
- Meghir, Costas, and Luigi Pistaferri. 2004. "Income Variance Dynamics and Heterogeneity." *Econometrica*, 72(1): 1–32.
- Michelacci, Claudio, Josep Pijoan-Mas, and Hernáan Ruffo. 2012. "Inequality in Unemployment Risk and in Wages." https://sites.google.com/site/hernanruffo/research.

- Mortensen, Dale. 2003. Wage Dispersion: Why are Similar Workers Paid Differently. MIT Press.
- Moscarini, Giuseppe, and Kaj Thomsson. 2007. "Occupation and Job Mobility in the US." *Scandinavian Journal of Economics*, 109(4): 807–836.
- Nágypal, Evá. 2005. "On the Extend of Job-to-Job Transitions." http://faculty.wcas.northwestern.edu/~een461/research.html.
- Nágypal, Evá. 2008. "Worker Reallocation Over the Business Cycle: The Importance of Employer-to-Employer Transitions." http://faculty.wcas.northwestern.edu/~een461/research.html mimeo.
- **Ortego-Marti, Victor.** 2012. "Unemployment History and Frictional Wage Dispersion." http://faculty.ucr.edu/~victorom/.
- **Papp, Tamás.** 2013. "Frictional Wage Dispersion with Bertrand Competition: An Assessment." *Review of Economic Dynamics*, 16(3): 540–552.
- **Postel-Vinay, Fabien, and Jean-Marc Robin.** 2002. "Equilibrium Wage Dispersion with Worker and Employer Heterogeneity." *Econometrica*, 70(6): 2295–2350.
- Shimer, Robert. 2005. "The Cyclical Behavior of Equilibrium Unemployment and Vacancies." *American Economic Review*, 95(1): 25–49.
- Siegel, Jeremy. 2002. Stocks for the Long Run. McGraw-Hill.
- Storesletten, Kjetil, Christopher I. Telmer, and Amir Yaron. 2004. "Consumption and Risk Sharing over the Life Cycle." Journal of Monetary Economics, 51(2): 609–633.
- Tjaden, Volker, and Felix Wellschmied. 2012. "Exploring the Causes of Frictional Wage Inequality." Institute for the Study of Labor (IZA) IZA Discussion Papers 6299.
- **Topel, Robert H.** 1991. "Specific Capital, Mobility, and Wages: Wages Rise with Job Seniority." *Journal of Political Economy*, 99(1): 145–176.
- Topel, Robert H, and Michael P Ward. 1992. "Job Mobility and the Careers of Young Men." The Quarterly Journal of Economics, 107(2): 439–79.